Author's response to reviews

Title: Essential health information available for India in the public domain on the internet

Authors:

Magdalena Z Raban (mrab2647@mail.usyd.edu.au)
Rakhi Dandona (rdandona@med.usyd.edu.au)
Lalit Dandona (dandona@u.washington.edu)

Version: 7 Date: 22 June 2009

Author's response to reviews: see over
22 June 2009

Editor
*BMC Public Health*

RE: Manuscript 4852768262299956

Dear Editor,

We have made all suggested formatting revisions in our manuscript except one with which we do not agree. Due to the nature of the data in Tables 2 and 3, these tables are longer than one page. These tables cannot be compressed into one page each, and these tables cannot be considered additional files as they must be part of the main paper because they contain our main findings. It should be possible to have tables of more than one page formatted for inclusion in the paper, as is done by many other journals. We would therefore appreciate the necessary arrangements by BMC Public Health to include these tables in the main body of the paper.

Yours sincerely,

Magdalena Raban
Rakhi Dandona
Lalit Dandona